Author’s response to reviews

Title: A bleeding colonic ulcer from invasive aspergillus infection in an immunocompromised patient: a case report

Authors:

Jorge Bizet (jorge.bizet@ttuhsc.edu)
Chad J. Cooper (chad.cooper@ttuhsc.edu)
Marc J. Zuckerman (Marc.Zuckerman@ttuhsc.edu)
Alireza Torabi (alireza.torabi@ttuhsc.edu)
Antonio Mendoza-Ladd (antonio.mendoza-ladd@ttuhsc.edu)

Version: 4 Date: 13 October 2014

Author’s response to reviews: see over
Dear Editor of *Journal of Medical Case Reports*,

It is our pleasure to submit our manuscript titled “Colonic Ulcer Bleeding in the Setting of Invasive Aspergillus Infection in an Immunocompromised Patient”. The manuscript has not been submitted for publication nor has it been published in whole or in part elsewhere. I attest to the fact that all authors listed on the title page have read the manuscript, attest to the validity and legitimacy of the data and its interpretation, and agree to its submission to the Journal of Medical Case Reports. All authors disclosed no conflict of interest pertaining to this publication. We feel that the findings described in our case report will be of special interest to the readers of Journal of Medical Case Reports and enhance the clinical and endoscopic practice of gastroenterology for patients with gastrointestinal bleeding.

Sincerely,

Antonio Mendoza-Ladd, MD
Texas Tech University Health Sciences Center
Department of Internal Medicine
4800 Alberta Avenue | El Paso, Texas 79905-2700
Email: antonio.mendoza-ladd@ttuhsed.edu